Author's response to reviews

Title: The pretreatment albumin to globulin ratio predicts chemotherapeutic outcomes in patients with unresectable metastatic colorectal cancer

Authors:

Masatsune Shibutani (fbxbj429@ybb.ne.jp)
Kiyoshi Maeda (tora@med.osaka-cu.ac.jp)
Hisashi Nagahara (hisashi@med.osaka-cu.ac.jp)
Hiroshi Ohtani (m5051923@msic.med.osaka-cu.ac.jp)
Yasuhito Iseki (vatchan_0622.ej20@me.com)
Tetsuro Ikeya (tetsu_47@msn.com)
Kenji Sugano (suganononono@yahoo.co.jp)
Kosei Hirakawa (hirakawa@med.osaka-cu.ac.jp)

Version: 5 Date: 15 April 2015

Author's response to reviews:

Response to EDITORS COMMENTS

1. In conclusion, we changed ‘The pretreatment AGR is a useful prognostic marker’ to ‘The pretreatment AGR may be a useful prognostic marker’.

2. We removed the abbreviations as much as possible.